Author's response to reviews

Title: Empowering patients through eHealth: A case report of a pan-European project

Authors:

Emanuele Lettieri (Emanuele.lettieri@polimi.it)
Lia P Fumagalli (liapaola.fumagalli@polimi.it)
Giovanni Radaelli (Giovanni.Radaelli@wbs.ac.uk)
Paolo Bertele’ (paolo.bertele@polimi.it)
Jess Vogt (jess.vogt@empirica.com)
Reinhard Hammerschmidt (reinhard.hammerschmidt@empirica.com)
Juan L Lara (juanl.lara.exts@juntadeandalucia.es)
Ana Carriazo (anam.carriazo@juntadeandalucia.es)
Cristina Masella (cristina.masella@polimi.it)

Version: 5
Date: 29 July 2015

Author's response to reviews: see over
Response to Editor

Dear Editor,

We are very grateful for your prompt reply and for having accepted our manuscript. We are honored to contribute to the BMC Health Services Research journal.

Following your final minor formatting requests, we implemented all them in a new version of the manuscript.

Therefore, we have made the following changes:

- The manuscript is now positioned as “debate” article instead of “case report” article;
- The main headings of both the abstract and the manuscript have been modified according to the guidelines for a “debate article”. In this view, the main headings, “Background, Case presentation and Conclusions”, have been changed in “Background, Discussion and Summary”

Please find attached in the present document the specific replies to the Editor.

We look forward to hearing from you.

With best regards,

The Authors
REPLIES TO EDITOR

We thank the Editor for the valuable comments. We addressed all the comments, as detailed below.

<table>
<thead>
<tr>
<th>EDITOR</th>
<th>REPLY</th>
</tr>
</thead>
<tbody>
<tr>
<td>Change the type of the article from “case report” article to “debate” article.</td>
<td>Accomplished request.</td>
</tr>
<tr>
<td>Change the main manuscript section headings to Background, Discussion, Summary.</td>
<td>Accomplished request. Please referred to lines 57, 116 and 417.</td>
</tr>
<tr>
<td>Change the abstract section headings to Background, Discussion, Summary (in the manuscript and the submission system).</td>
<td>Accomplished request. In the manuscript, please referred to lines 28, 38 and 50.</td>
</tr>
</tbody>
</table>